WEDNESDAY

Cover Page

Computer Science CSC 364H Summer 2002 St. George Campus University of Toronto Homework Assignment #

Please fill out this page and attach it to the front of your assignment. Without this completed cover page, your assignment will not be marked.

First Name: Last Name:

Student Number:

Tutor (circle one): Spyros Angelopoulos Ying Zhu Babak Farzad A–G H–M N–Z

“I have read and understood the policy on Collaboration in Homework stated in the Course Information handout. I assert this work is my own, and I appropriately acknowledge any and all sources I used as a source for ideas.”

Signature:

(Without your signature, your assignment will not be marked.)

Acknowledgements: (You need not acknowledge the course textbook or notes.)